Author's response to reviews

Title: Polymorphism in COX-2 modifies the inverse association between Helicobacter pylori seropositivity and esophageal squamous cell carcinoma risk in Taiwan: a case control study

Authors:

Huang-Ming Hu (huhuangming@ms66.url.com.tw)
Chao-Hung Kuo (kjh88@seed.net.tw)
Chien-Hung Lee (cnhung@kmu.edu.tw)
I-Chen Wu (minicawu@yahoo.com.tw)
Ka-Wo Lee (kawolee@kmu.edu.tw)
Jang-Ming Lee (jangming@ha.mc.ntu.edu.tw)
Yih-Gang Goan (goan@ksts.seed.net.tw)
Shah-Hwa Chou (0740099@kmhk.kmu.edu.tw)
Ein-Long Kao (kaeil@cc.kmu.edu.tw)
Ming-Tsang Wu (e_encourage@yahoo.com)
Deng-Chyang Wu Deng-Chyang Wu (dechw@yahoo.com)

Version: 3 Date: 14 May 2009

Author's response to reviews: see over
Cover letter

Title: Polymorphism in COX-2 modifies the inverse association between Helicobacter pylori seropositivity and esophageal squamous cell carcinoma risk in Taiwan: a case control study

Authors:
Huang-Ming Hu  hhuangming@ms66.url.com.tw
Chao-Hung Kuo  kjh88@seed.net.tw
Chien-Hung Lee  cnhung@kmu.edu.tw
I-Chen Wu  minicawu@yahoo.com.tw
Ka-Wo Lee  kawolee@kmu.edu.tw
Jang-Ming Lee  jangming@ha.mc.ntu.edu.tw
Yih-Gang Goan  goan@ksts.seed.net.tw
Shah-Hwa Chou  0740099@kmhk.kmu.edu.tw
Ein-Long Kao  kaeil@cc.kmu.edu.tw
Ming-Tsang Wu  e_encourage@yahoo.com
Deng-Chyang Wu  dechwu@yahoo.com

Dear Editor:
We did the following change according your suggestion.
1. Reference 35 - please provide full details for this reference.
   Answer: We have added more information about the software we used in reference 35. Please see page 21.
2. Table size (see Tables 2, 3 and 4) - These must fit into an A4 portrait page. If you cannot re-orientate the table onto a single a4 portrait page (maximum size 176mm x 230mm), please move it to the additional files.
   Answer: We have modified Table 2,3 and 4 into A4 portrait page. Please see page 26 27 and 28.
3. Abstract - please remove keywords from this section.
   Answer: we did that following your request.
4. Authors' contributions - please use the following format (please use initials to refer to each author's contribution)
   Answer: We have change the format in this section. Please see page 17.
5. List of abbreviations - please rename `Abbreviations'.
   Answer: we did that following your request.

With kind Regards
Huang-Ming Hu